LHCb Computing Operations

Thursday, April 19, 2007

Present: Joel Closier, Davide Perego, Andrew Smith, Marcos Seco Miguelez, Gianluca Castellani, Roberto Santinelli, Raja Nandakumar, Gennady Kuznetsov, Cedric Poterat, Ricardo Graciani-Diaz, Marianne Bargiotti, Stuart Paterson, Nick Brook, Philippe Charpentier

Simulation:

Every request has been fulfilled.

Reconstruction:

After the cleaning of the processing DB and the checking of the LFC catalog, we have started to populate the processing DB with Maxi Bias information:

PIC: problem with the TURL which is return after the modification made within SRMv1r1. ROOT is not able to access the data with rfio:///stage/server/lh/xxxx.1234

RAL: no pilot agent are running because of the fairshare

NIKHEF: problem of software installation (after the meeting we found that raja credentials was mapped to lhcbsgm)

CERN : OK IN2P3 : OK

Gridka: no jobs was submitted yet

Stripping:

We have problem with the options used inside DaVinci and Brunel but other wise in term of DIRAC, it is ok.

AOB:

LCG MB: TOP 5 issues:

- 1 Data access (Roberto)
- 2 glExec (Stuart)
- 3 File management (Andrew)
- 4 Application deployement and release procedure .. (Nick)
- 5 Information system, BDII, ...(Roberto)
- 1 slide for each part should provided for Monday and it will be discussed during the PASTE meeting.